Author’s response to reviews

Title: High Prevalence of Substance Use among High School Adolescents in Woreta Town, Northwest Ethiopia and Associated Factors: Multi-Domain Factor Analysis

Authors:

Anteneh Messele (antenehmessele33@gmail.com)
Telake Azale (telakea@yahoo.com)
Solomon Meseret (messisol@gmail.com)

Version: 7 Date: 4 November 2014

Author’s response to reviews:

Dear Editors,

Warm Greetings,

We would like to thank you for follow-up and assistance.

Your request to include the ethics and consent explanation into the method section of the manuscript is accepted. The changes made in the manuscript are noted here.

1. Name of the grandfather of the authors is included now

2. Here is the point included in as an explanation “There were high proportions of participants coming from the rural areas (45.6%) whose families are in most cases illiterate (37%). High school students in the study site were older than their counterparts. This was shown with the average age of the participants was 17.25 years (SD=1.24). This is nearly 18 years. These students are competent /capable to give informed consent. The study is also a descriptive study with no known risk to the study participants. The research ethics committee gave us the ethical clearance considering the mention issues in the local context.”

We thank you so much.

With Kindest Regards,
Anteneh Messele

Corresponding Author